Author's response to reviews

Title: Interactions between Social/ behavioral Factors and ADRB2 Genotypes May be Associated with Health at Advanced Ages in China

Authors:

Yi Zeng (zengyi68@gmail.com)
Lingguo Cheng (chenglingguo@gmail.com)
Ling Zhao (ljzhao@ucsd.edu)
Qihua Tan (qtan@health.sdu.dk)
Qiushi Feng (qiushi_feng@hotmail.com)
Huashuai Chen (huashuai.chen@gmail.com)
Ke Shen (shenke.ccerpku@yahoo.com.cn)
Jianxin Li (ljx@pku.edu.cn)
Fengyu Zhang (zhangf28@gmail.com)
Huiqing Cao (caohuiqing@pku.edu.cn)
Simon G. Gregory (simon.gregory@duke.edu)
Ze Yang (yangze016@yahoo.com.cn)
Jun Gu (gj@pku.edu.cn)
Wei Tao (weitao@pku.edu.cn)
Xiao-Li Tian (tianxiaoli@pku.edu.cn)
Elizabeth R. Hauser (Elizabeth.Hauser@duke.edu)

Version: 2 Date: 24 May 2013

Author's response to reviews: see over
Dear editors of BMC Geriatrics,

On behalf of my co-authors, we are re-submitting our substantially revised and improved manuscript entitled “Interactions between Social/behavioral Factors and ADRB2 Genotypes May be Associated with Health at Advanced Ages in China”, based on the very thoughtful comments provided by the Section Editor and the two reviewers. The significantly revised parts are marked in blue-color. The interesting new findings reported in this article have led us to believe that it may be very interested to various readers of BMC Geriatrics.

We would greatly appreciate your consideration for publication of our paper at BMC Geriatrics. If you have any questions or concerns, please let me know. Thank you very much for your consideration.

Sincerely yours,

Yi Zeng, Ph.D.
Professor, Center for Study of Aging and Human Development and Geriatric Division of Medical School; Director of Center for Chinese Populations and Socioeconomic Studies, Duke University
Professor and Director of Center for Healthy Aging and Development Studies, National School of Development, Peking University
Distinguished Research Scholar, Max Planck Institute for Demographic Research
Foreign member of the Royal Netherlands Academy of Arts and Sciences